Author's response to reviews

Title: Acute effect of meal glycemic index and glycemic load on blood glucose and insulin responses in humans

Authors:

Jose J Galgani (jgalgani@inta.cl)
Carolina C Aguirre (caguirre@inta.cl)
Erik E Diaz (ediaz@inta.cl)

Version: 3 Date: 20 June 2006

Author’s response to reviews:

Dear Editor,

We have done all corrections asked in your last e-mail, which are detailed:
1. English translation by a native speaker.
2. It is explained the meaning of 23 and 49%.
3. It is changed * by the suggested symbol.
4. It is corrected the legend for figure 2.
5. It is removed the unnecessary capitalization.